Author's response to reviews

Title: The effectiveness of case management for comorbid diabetes type 2 patients; the CasCo study. Design of a randomized controlled trial.

Authors:

Nathalie Versnel (n.versnel@nivel.nl)
Laura MC Welschen (l.welschen@vumc.nl)
Caroline A Baan (caroline.baan@rivm.nl)
Giel Nijpels (g.nijpels@vumc.nl)
François G Schellevis (f.schellevis@nivel.nl)

Version: 2 Date: 8 June 2011

Author's response to reviews:

Dear editor,

There has been a revision made on the front page of the manuscript. The name of one of the co-authors (François G Schellevis) was missing, only the email address was visible.

We look forward to receiving the editorial decision concerning the manuscript.

On behalf of all authors,

Yours sincerely,

Nathalie Versnel, MSc

Nathalie Versnel